Author's response to reviews

Title: Lessons from Mycobacterium avium complex associated pneumonitis: a case report

Authors:

Victor Zota (zotav@ummc.org)
Sheryn M Angelis (angeliss@ummc.org)
Armando E Fraire (frairea@ummc.org)
Ciaran McNamee (mcnameec@ummc.org)
Shasta Kielbasa (shasta.kielbasa@umassmed.edu)
Daniel H Libraty (daniel.libraty@umassmed.edu)

Version: 10 Date: 28 April 2008

Author's response to reviews:

The following revisions have been made:

1. Authors' contributions- statement added "All authors read and approved the final manuscript"

2. Corresponding author- email address listed

3. Sections after Conclusions re-ordered as requested.